Author’s response to reviews

Title: Composition and Diversity of the Subgingival Microbiome and its Relationship with Age in Postmenopausal Women: an Epidemiologic Investigation

Authors:

Michael LaMonte (mlamonte@buffalo.edu)
Robert Genco (rjgenco@buffalo.edu)
Michael Buck (mjbuck@buffalo.edu)
Daniel McSkimming (dim@buffalo.edu)
Lu Li (lli59@buffalo.edu)
Kathleen Hovey (koreilly@buffalo.edu)
Christopher Andrews (candrews@buffalo.edu)
wei Zheng (zhengw@buffalo.edu)
Yijun Sun (yijunsun@buffalo.edu)
Amy Millen (aemillen@buffalo.edu)
Maria Tsompana (mtsompana@buffalo.edu)
Hailey Banack (hrbanack@buffalo.edu)
Jean Wactawski-Wende (jww@buffalo.edu)

Version: 2 Date: 28 Aug 2019

Author’s response to reviews:

A detailed response to the reviewer comments has been uploaded as a Word doc.